Impact of line probe assay-based molecular testing on individualized treatment in patients with rifampicin-resistant tuberculosis: data from the prospective **INNOVA4TB** cohort study in Ukraine

Andrii Dudnyk^(D), Matthias Hempel, Oksana Lytvyniuk, Halyna Liudkevych, Volodymyr Matsera, Tetiana Nikitchenko, Svitlana Blyzniuk, Barbara Molina-Moya, **Rosemarie Preyer and José Domínguez**

Abstract

Background: Ukraine remains a high World Health Organization priority country for drugresistant tuberculosis (TB). Rifampicin-resistant TB (RR-TB) has a more protracted, more complicated, and more expensive treatment. In 2021, Ukraine reported 4025 RR-TB cases -5.4 times more (751) than all 30 European Union/ European Economic Area countries together. **Objectives:** The objective of the study was to determine the diagnostic accuracy of line probe assay (LPA), AID Autoimmun Diagnostika GmbH, for detecting resistance to anti-TB drugs and its clinical application for selecting treatment regimens.

Design: A prospective observational cohort study.

Methods: From May 2019 to June 2020, we consecutively enrolled patients with active TB hospitalized at the Regional Phthisiopulmonology Center (Vinnytsia, Ukraine), aged between 18 and 82 years. The LPA was performed in the Genetic Research Laboratory at National Pirogov Memorial Medical University, Vinnytsia, Ukraine.

Results: A total of 84 clinical specimens and 97 culture isolates from 126 TB patients were tested during the study. Accuracy (95% confidence interval) of LPA for clinical samples in comparison with phenotypic drug susceptibility test (DST) was 80.1 (68.5-89.0) for isoniazid (H), 74.7 (62.4–84.6) for rifampicin (R), 74.4 (62.5–84.1) for ethambutol, 71.4 (41.9–91.6) for streptomycin, 84.6 (62.4-96.5) for prothionamide/ethionamide, and 84.6 (73.6-92.3) for levofloxacin (Lfx), respectively. We found a significantly higher sensitivity of LPA for H, R, and Lfx for the culture isolates compared to clinical specimens (p < 0.05). LPA detected different mutations in 6 out of 17 (35.5%) patients susceptible to R by Xpert. A shorter treatment regimen with an injectable agent demonstrated a low suitability rate of 5% (8/156) in a cohort of RR-TB patients from Ukraine.

Conclusion: Initial LPA testing accurately identifies resistance to anti-TB drugs and facilitates the selection of an appropriate treatment regimen, minimizing exposure to empirical therapy.

Ther Adv Respir Dis

2024. Vol. 18: 1-18 DOI: 10 1177/ 17534666241249841

© The Author(s), 2024.

Article reuse auidelines: sagepub.com/journalspermissions

Correspondence to: Andrii Dudnyk

Department of Tuberculosis. Clinical Immunology and Allergy, National Pirogov Memorial Medical University, 56 Pyrogova St., Vinnytsia 21018. Ukraine

Institut d'Investigació Germans Trias i Pujol (IGTP), Mar Building, P1-18, Carretera de Can Ruti, Camí de les Escoles s/n. Badalona 08916. Barcelona, Spain

adudnyk@igtp.cat

Matthias Hempel **Rosemarie Prever** AID/GenID GmbH. Straßberg, Germany

Oksana Lytvyniuk Department of Tuberculosis, Clinical

Immunology and Allergy National Pirogov Memorial Medical University, Vinnytsia, Ukraine

Halyna Liudkevych

Genetic Research Laboratory, National Pirogov Memorial Medical University, Vinnytsia, Ukraine

Volodymyr Matsera Tetiana Nikitchenko

Department of Drug-Resistant Tuberculosis, Regional Phthisiopulmonology Center, Bohonyky, Vinnytsia Region, Ukraine

journals.sagepub.com/home/tar



Svitlana Blyzniuk Department of Drug-Susceptible Tuberculosis, Regional Phthisiopulmonology Center, Bohonyky, Vinnytsia Region, Ukraine

Barbara Molina-Moya Institut d'Investigació Germans Trias i Pujol (IGTP), Badalona, Barcelona, Spain

José Domínguez Institut d'Investigació Germans Trias i Pujol (IGTP), Badalona, Barcelona, Spain

Department of Genetics and Microbiology, Universitat Autònoma de Barcelona, Badalona, Barcelona, Spain

CIBER Enfermedades Respiratorias, Badalona, Barcelona, Spain

Plain language summary

Study about the impact of rapid resistance detection on the treatment of patients with tuberculosis in Ukraine written by healthcare and biomedical professionals to better understand how we can improve the results of treatment and to prevent spreading of resistant bacteria

Why was the study done? Ukraine has over 4000 patients with tuberculosis (TB) resistant to at least one drug (rifampicin) - five times that of all 30 European Union/ European Economic Area countries combined. Unfortunately, only about 60% of such patients have been successfully treated in 2019. At that time, the majority of people suffering from tuberculosis in Ukraine, after checking resistance to rifampicin, initially received standard combinations of the first-line or second-line anti-TB medicines before the result of traditionally used tests (usually few weeks later) became available to individualize the treatment. Alternatively, the sputum could be transported to some overloaded reference laboratories located hundreds of km away from the treatment places.

What did the researchers do? The INNOVA4TB team implemented rapid diagnostics of drug resistance in routine practice, guiding key antibiotics use in TB patients. A total of 181 samples from 126 individuals were tested during 2019-2020.

What did the researchers find? This new diagnostic technology accurately detected resistance to 9 anti-TB drugs in sputum samples. It could be helpful to select appropriate TB treatment regimens, reducing time for decision from 1 month up to 2 days. Recommended at the study time 9-month shorter standardized treatment regimen with injectable agent was suitable only for 5% of patients for whom it was indicated in Vinnytsia region of Ukraine.

What do the findings mean? The study has demonstrated successful implementation of the new molecular diagnostic technology from scratch in a country with restricted resources and limited TB laboratory capacity. This test can facilitate optimal distribution of available wards among patients with different profiles of resistance and correct choice between treatment options.

Keywords: drug-resistant tuberculosis, line probe assay, molecular diagnostics, TB treatment

Received: 12 December 2023; revised manuscript accepted: 10 April 2024.

Introduction

Tuberculosis (TB) remains the deadliest infectious disease in many developing countries,¹ despite being preventable and curable. Recent global progress in the fight against TB has been due to the accelerating implementation of rapid diagnostic methods,² new anti-TB drugs, and promising shorter treatment regimens.³ However, first the COVID-19 pandemic and then the war in Ukraine have reversed these advances. Nowadays, Europe faces one of the biggest waves of migration in history, hosting almost 8 million Ukrainians while there are an estimated 6.5 million internally displaced people within Ukraine.⁴ Undoubtedly, the destruction of healthcare facilities throughout Ukraine, combined with war-initiated humanitarian and economic disasters, will have a cumulative effect on TB prevalence beyond Ukraine itself.⁵ To prevent new TB outbreaks in these circumstances, it is essential to develop laboratory capacity for accurate TB diagnosis and timely detection of drug resistance. Multidrug-resistant tuberculosis (MDR-TB) is defined as having resistance documented by a drug susceptibility test (DST) to at least isoniazid (H) and rifampicin (R). In contrast, extensively drug-resistant TB (XDR-TB) is defined as additional resistance of Mycobacterium tuberculosis to fluoroquinolones (FO) and at least either linezolid (Lzd) or bedaquiline (Bdq).⁶ Although monoresistance to R is a rare phenomenon, we also operate with the term rifampicin-resistant TB (RR-TB) in situations when DST to other drugs was not performed or when results are still pending. The previously widely used second-line injectables (SLIs) capreomycin (Cm) and kanamycin (Km) are no longer recommended for the treatment of M/XDR-TB due to an increased risk of unfavorable treatment outcomes.⁷ Currently, only two aminoglycosides, amikacin (Am) and streptomycin (S) can be administered to a limited number of patients with M/XDR-TB who cannot be treated with entirely oral regimens.

Ukraine is ranked as one of the countries with the highest M/XDR-TB burdens.1 In 2021, there were reported 4025 RR-TB cases - in 5.4 times more (751) than all 30 European Union/European Economic Area countries together.8 In 2022, Ukraine's RR-TB incidence was firstly increased since many years corresponding to 9.5 per 100,000 population.9 However, patients with TB have limited access to next-generation sequencing for a DST, and only 4 out of 34 (12%) third-level TB microbiological laboratories have World Health Organization (WHO)-recommended line probe assay (LPA) technology GenoType MTBDRsl (Hain Lifescience GmbH, Nehren, Germany). Shipment of clinical samples to reference laboratories with LPA delays test results, particularly during irregular supply of reagents and consumables. Since 2021, the GeneXpert MTB/XDR cartridges (Cepheid, Sunnyvale, CA, USA) have been used in Ukraine for the prediction of resistance to H, FQ [levofloxacin (Lfx) and moxifloxacin (Mfx)], Am, and prothionamide/ethionamide (Pt/Et). However, programmatic assessment of this assay in routine practice and comparison with LPA has yet to be performed in Ukraine.

Conventional methods for culture isolation of M. *tuberculosis* from clinical samples and phenotypic DST (pDST) using critical concentrations (CC) to anti-TB drugs have a long turnaround time (TAT). It is worth mentioning that patients with rpoB mutations detected by Xpert MTB/RIF usually shared ward space with other individuals with pending RR-TB results for pDST to other anti-TB drugs. With growing knowledge about genetic mutations,¹⁰ and the development of the WHO Consolidated Guidelines on Tuberculosis¹¹ describing clinical interpretation of molecular assay results, the genetic DST (gDST) was intended as an initial test for aiding treatment selection. Several studies analyzed the specificity and sensitivity of LPA in clinical samples,12-15 but the clinical utility of this diagnostic tool for recruitment of patients for a 9-11-month short treatment regimen with an injectable agent (4-6Km-Mfx-Pt-Cfz-Z-H_{hd}-E/5Mfx-Cfz-Z-E) as was recommended by the WHO¹⁶ is still unknown.

For this purpose, we have conducted a prospective observational cohort study (ClinicalTrials.gov Identifier: NCT04002869) with the primary objective of evaluating the diagnostic accuracy of LPA (AID Autoimmun Diagnostika GmbH, Strassberg, Germany) for detecting resistance to anti-TB drugs. We have performed a large-scale within-study comparison of LPA testing on clinical specimens and culture isolates from the same cohort of patients. The second objective of our study was to analyze the clinical utility of initial LPA testing as a decision-making tool to select treatment regimens for patients with RR-TB according to the current WHO recommendations for clinical practice at the time of our study, which are individualized longer 18-20-month regimens consisting of old or new/repurposed medicines17 and standardized shorter (9-11 month) regimens with an injectable agent.18 We also studied whether the simultaneous presence of rpoB, katG, and inhA mutations in M. tuberculosis isolates could be associated with FQ resistance in our cohort.19

Methods

Study design and setting

This prospective observational cohort study evaluated the diagnostic accuracy of the LPA testing of 181 samples (84 clinical specimens and 97 culture isolates) compared to a reference standard (pDST) at the Genetic Research Laboratory at National Pirogov Memorial Medical University (Vinnytsia, Ukraine). The study results were reported by the 'Standards for Reporting Diagnostic Accuracy Studies'²⁰ (see Supplemental Checklist).

Study population

From May 2019 to June 2020, we enrolled patients with active TB hospitalized at a TB dispensary (Vinnytsia, Ukraine) aged between 18 and 82 years. The main eligibility criteria were a positive sputum Xpert test with *M. tuberculosis* complex (MTBC) DNA detected and voluntary willingness to participate in the study. The patients were consecutively recruited into the study until the desired sample size was reached. Individuals with a negative culture result and no pDST for comparison with LPA were retrospectively excluded from the analysis.

Study procedures

Morning sputum specimens were collected in an isolated room following infection control precautions and were sent to the laboratory for further processing. Sputum collection was performed within the first week after admission to minimize the effect of treatment on bacterial load and cough productivity. If the person had been on treatment for more than 7 days before agreeing to participate in the study, or we could not organize sputum collection, then only culture isolate, not a direct clinical sample, was collected for LPA testing. Only spontaneously expectorated sputum samples were collected. However, one pleural fluid specimen was also included. A standard procedure using N-acetyl-L-cysteine and NaOH was used for decontamination and homogenization of sputum specimens before they were routinely analyzed by smear microscopy, culture on solid medium Löwenstein-Jensen and liquid medium with BACTEC Mycobacteria Growth Indicator Tube (MGIT) 960 system (Becton Dickinson Diagnostic Systems, Sparks, MD, USA), and by Xpert MTB/RIF platform (Cepheid). LPA analysis was performed on the clinical isolates and on the decontaminated sputum samples. Phenotypical DST based on CC was performed on positive cultures following the WHO guidelines and national protocols.²¹ The National ΤB Reference Laboratory performed regular external quality controls on the regional microbiology laboratory.

LPAs are reverse-hybridization DNA strip-based tests involving the following consecutive steps:

DNA extraction from biological sample, the polymerase chain reaction (PCR) amplification followed by reverse hybridization with sequencespecific oligonucleotide probes and visual analysis of the nitrocellulose strip recording detection of signals of wild-type (WT) probes and/or probes corresponding to the most common mutations (MUT). Each clinical specimen (irrespective of smear microscopy status) and culture isolate were tested for resistance to first- and second-line drugs. All steps of LPA testing were performed following the manufacturer's instructions and supervised by a staff member with appropriate skills in molecular assays. All staff involved in the molecular testing underwent training in conducting LPA procedures.

DNA isolation. Mycobacterial DNA extraction was done from 500 µL aliquot of each specimen using commercially available extraction kits, QIAamp DNA Blood Mini Kit (Qiagen, Hilden, Germany), and AID Smart Lysis Tool (AID/GenID GmbH, Straßberg, Germany) for culture isolates and decontaminated sputum samples, respectively.

Multiplex PCR amplification and reverse hybridization. A multiplex PCR was performed using DNA isolates with specific primers and nucleotide mixes for amplification in each diagnostic module. Following PCR, the respective biotinylated amplicons were identified by a hybridization reaction with oligonucleotide probes corresponding to a specific wild-type or/and mutation at nucleic acid positions and controls immobilized on a nitrocellulose membrane in a distinctive line format. Due to washing stage, the hybrids between amplicons and gene probes are only immobilized if the sequence of the probe is totally complementary to amplified DNA. The color reaction, as a result of binding of streptavidin-coupled alkaline phosphatase with hybrids, was then visually detected and interpreted.

Three commercially available LPA modules (strips) manufactured by AID Autoimmun Diagnostika GmbH to identify MTBC in clinical samples and its resistance to the main anti-TB drugs. Every module includes conjugate, amplification (glyceraldehyde-3-phosphate dehydrogenase gene for DNA quantity and quality), *Mycobacterium* universal, and MTBC controls. *M. tuberculosis* strain H37Rv and PCR-grade water were used as a positive and negative control samples, respectively.

The first module (MDR-TB) is used to detect WT and specific mutations associated with H (*katG* and *inhA* promoter), R (*rpoB*), and Pt/Et (*inhA* promoter) resistance. The second module (FQ-E) is used to detect WT and specific mutations associated with Lfx and Mfx (*gyrA*), and ethambutol (E) (*embB*) resistance. The third module (SLIs) is used to detect WT and specific mutations associated with aminoglycosides Am, Km, (*rrs*), S (*rpsL* and *rrs*), and polypeptide Cm (rrs) resistance.

Interpretation. Following the interpretation and reporting guide for laboratory staff and clinicians,²² we used the term 'Resistance not detected' instead of 'Susceptible,' taking into consideration that even the presence of all WT probes cannot rule out resistance to the corresponding drug. Resistance was 'inferred' when one or more WT probes were not observed and none of the mutation probes in the corresponding region were developed. Resistance was 'detected' when any mutation probes identifying specific mutations conferring resistance to the drugs were bound with the amplicons (regardless of whether WT probes were developed). The test was considered to be invalid when the conjugate, Mycobacterium universal, or MTBC control did not produce any visible signal or there were issues with positive or/and negative control samples.

Clinical information and reference standard results were unavailable to the index test performers. However, routinely collected clinical information was available for the assessors of the reference standard, but we didn't provide them with the results of index test.

Assessing impact of molecular methods on TB treatment

Routinely collected data on the treatment regimens administered to the patients were analyzed. After obtaining pDST results, we traced any changes in the treatment regimen based on Xpert testing. Then, we evaluated the feasibility and acceptability of a shorter RR/MDR-TB treatment with injectable agent in our cohort of patients and how initial LPA testing would contribute to the selection of appropriate treatment regimen if implemented.

Data analysis

We calculated sensitivity, specificity, and predictive values for diagnostic accuracy assessments. The calculations were based on pDST results, where we used MGIT data as the preferred comparator whenever possible because it has CC defined for most of the drugs. Indeterminate LPA and pDST results were reported to estimate the impact on routine practice but were excluded from the analysis. We performed a blind interpretation of LPA testing results to minimize the effect of cognitive bias in the analysis. Missing data on the LPA and reference standard were also discarded from further evaluation. Although LPA results can take more than two values (resistance detected, inferred, and not detected), categorization of test results as positive or negative for sensitivity and specificity calculation was based on the presence of the mutation probe strip: binding of the amplicons to MUT probes was reported as resistance detected, and both, resistances inferred (no WT band developed) and identifying any of the WT probes without mutation bands at the same gene target were reported as resistance not detected.22

For sensitivity, specificity, accuracy, and predictive values, the MedCalc Software Ltd. (Ostend, Belgium) Diagnostic test evaluation calculator²³ was used. The proportions were compared using Ζ Score Calculator for 2 Population Proportions.²⁴ Statistical analysis and graphics were performed with R Statistical Software [v4.2.3; R Core Team (15 March 2023)]. We used a calculator simulating sampling for the fixed pool size and assumed known test sensitivity and specificity for initial sample size estimation. Intended sample size estimation was calculated online (http://wnarifin.github.io), reflecting national surveillance data²⁵ of drug resistance to the targeted anti-TB medicines at 30%, expected overall LPA sensitivity of 0.8 and specificity of 0.9. The final estimated minimal sample size (5% dropout), ndrop = 123.

Results

Clinical evaluation of LPA modules

Eighty-four clinical specimens and 97 culture isolates from 126 active TB patients were tested in the study setting and included in the data analysis. The median age of patients was 43 years, interquartile range (IQR) 15 and 28 were assigned female at birth (22.2%). A flow diagram of participants who passed the screening for enrolment into the study and those who have withdrawn from the analysis is shown in Figure 1.



Figure 1. Number of patients and samples initially enrolled and finally assessed in the study. GeneXpert MTB+Rif-, Mycobacterium tuberculosis complex detected but no resistance to rifampicin identified; GeneXpert MTB+Rif+, Mycobacterium tuberculosis complex, and resistance to rifampicin detected; LJ&MGIT, phenotypic culture isolates form Löwenstein–Jensen solid medium or from Mycobacteria Growth Indicator Tube; LPA, line probe assay.

LPA specificity (%) in detecting RR-TB by testing clinical specimens was 80.0 (95% CI 51.9-95.7), while sensitivity (%) was only 64.7 (95%

regimen for patients with R-susceptible TB, it is essential to provide upfront detection of resistance to H (Hr-TB). In our study, the sensitivity CI 50.1–77.6). Then, to design the treatment to identify resistance to H by testing clinical

specimens was 80.4 (95% CI 66.9-90.2), and the specificity - 80.0 (95% CI 51.9-95.7). The rifapentine (P) based regimen for drug-susceptible TB, both RR-TB and Hr-TB treatment regimens, requires FQ in the scheme. Thus such testing of clinical samples is a high priority for the management of TB. The best specificity was obtained for Lfx 93.0 (95% CI 80.9-98.5). Nevertheless, the sensitivity of LPA in the detection of FO resistance was low and varied from 8.7% (Lfx) to 12.5% for low-level resistance Mfx_{0.25} (critical concentration value of 0.25; Mfx could be considered as an effective drug only at a high dose up to 800 mg in shorter regimen if concomitant high-level resistance was not documented). Interestingly, using culture isolates instead of clinical samples improves LPA sensitivity in identifying resistance to Lfx - 6.7-fold increase (from 8.7% to 58.3%) and 5.7-fold increase to Mfx_{0.25} (from 12.5% to 71.4%). Estimates of diagnostic accuracy in comparison with pDST are represented in Table 1.

Within-study comparison of LPA accuracy in clinical samples versus culture isolates

For a direct within-study comparison of LPA between clinical specimens and culture isolates, we prospectively evaluated available testing results of both samples from 45 patients. Accuracy with corresponding 95% confidence intervals (CI) of LPA (pDST as a comparator test) was higher when performed in isolated strains than in clinical samples, except for FQ (Figure 2). However, a statistically significant difference was observed only for H – 100 (95% CI 89.7–100) *versus* 75.6 (95% CI 57.9–88.6) two-tailed *z* test, p=0.002. However, we found significantly higher sensitivity for H, R, and Lfx analyzing culture isolates (p < 0.05). More detailed information is presented in Table 2.

Second-line gDST in R-susceptible TB cases

Of particular interest is the rate of resistance to second-line drugs in patients with TB and documented susceptibility to R because these data are not routinely collected. As shown in Table 3, in 6/17 (35.3%) clinical samples and 14/35 (40.0%) culture isolates from patients without the *rpoB* mutation detected by Xpert, LPA identified mutations in other regions associated with resistance to some anti-TB drugs, including rifampicin.

Potential alternatives that might be considered in settings with no access to molecular detection of FQ resistance

In our study, 66 clinical samples were tested by LPA with FQ resistance documented by pDST. Among those specimens with no detected mutations associated with resistance to FO, over 90% (60/66) of the amplicons did not bind to WT probes, inferring resistance. In comparison, specific mutations were detected in 7.6% (5/66) of samples (Table 4). According to our data, based on 79 culture isolates with LPA testing, a combination of rpoB, katG, and inhA mutations was significantly associated with FO resistance detected by pDST $\gamma(1) = 7.4$ (Pearson Chi-Square test with one degree of freedom), p = 0.007. Individuals with such mutations are 3.8 times more likely to have FQ resistance than remaining patients [odds ratio 3.8 (95% CI 1.4-10.1)].

Selection of treatment regimen based on LPA results

To assess the clinical utility of LPA implementation in selecting treatment regimens (either standardized/short or individualized/long) in routine practice, we analyzed a cohort of all RR/MDR-TB patients hospitalized at our clinic in 2019. From 156 patients with RR/MDR-TB, only 35 (22%) patients met the criteria for prescribing a shorter treatment regimen with an injectable agent (4-6 Km/Am-Mfx-Pt/Et-Cfz-Z-H_{hd}-E/5 Mfx-Cfz-Z-E), such as no exposure to one or more secondline medicines in the regimen for >1 month; no suspected ineffectiveness to compound in the shorter MDR-TB regimen; no pregnancy; no bilateral disseminated lesions on the chest X-ray, or severe extrapulmonary TB (meningeal TB or TB spondylitis). All patients started the shorter regimen pending the results of the pDST.

Due to the detection of additional resistance by pDST (exclusion criteria from shorter treatment), 22/35 (63%) patients with RR-TB were transferred to individualized long-term treatment schemes, another 5/35 (14%) patients were excluded due to documented intolerance or toxicity associated with one or more agents in the shorter RR-TB regimen. Therefore, only 8/35 (23%) patients completed an initial shorter treatment regimen with an injectable agent, demonstrating a low suitability rate of 5% (8/156) of cases in the RR-TB cohort. For these 35 patients with RR-TB enrolled in the study, for 7 patients,

THERAPEUTIC ADVANCES in *Respiratory Disease*

Table 1. Results of LPA testing of clinical samples and ICs in comparison with culture-based drug susceptibility tests.

Anti-TB drug	LPA resistance detection						
	Type of sample	Sensitivity (%), (95% CI)	Specificity (%), (95% Cl)	PPVª (%) (95% CI)	NPVª (%) (95% CI)	Accuracyª (%) (95% Cl)	
First-line							
H katG inhA	CS	41/51 (80.4), (66.9–90.2)	12/15 (80.0), (51.9–95.7)	68.4 (43.8–85.7)	88.3 (80.5–93.3)	80.1 (68.5–89.0)	
	IC	70/71 (98.6), (92.4–100)	14/15 (93.3), (68.1–99.8)	88.8 (54.5-98.2)	99.2 (94.6-99.9)	95.2 (88.3–98.6)	
R rpoB	CS	33/51 (64.7), (50.1–77.6)	12/15 (80.0), (51.9–95.7)	63.5 (36.3–83.0)	80.8 (72.9–86.8)	74.7 (62.4–84.6)	
	IC	56/65 (86.2), (75.3–93.5)	20/21 (95.2), (76.2–99.9)	90.7 (58.9–98.5)	92.7 (87.4–95.9)	92.1 (84.2–96.8)	
E emb	CS	12/37 (32.4), (18.0–49.8)	32/33 (97.0), (84.2–99.9)	85.2 (44.2–97.7)	72.7 (67.9–77.1)	74.4 (62.5–84.1)	
	IC	23/43 (53.5), (37.7–68.8)	34/37 (91.9), (78.1–98.3)	78.0 (53.7–91.6)	78.6 (72.4–83.7)	78.5 (67.8–86.9)	
Second-line							
S rpsL rrs	CS	5/7 (71.4), (29.0–96.3)	5/7 (71.4), (29.0–96.3)	21.7 (7.3–49.5)	95.7 (86.4–98.8)	71.4 (41.9–91.6)	
	IC	14/14 (100), (76.8–100)	12/13 (92.3), (64.0–99.8)	59.1 (18.0-90.5)	100	93.1 (76.4–99.2)	
Km <i>rrs</i> 1401, 1484, 1402	CS	2/21 (9.5), (1.2–30.4)	37/39 (94.9), (82.7–99.4)	17.1 (3.0–57.7)	90.4 (89.0-91.7)	86.3 (75.0–93.8)	
	IC	9/27 (33.3), (16.5–54.0)	38/39 (97.4), (86.5–99.9)	59.1 (16.3–91.5)	92.9 (90.9–94.5)	91.0 (81.4–96.7)	
Cm <i>rrs</i> 1401, 1484	CS	2/14 (14.3), (1.8–42.8)	41/43 (95.4), (84.2–99.4)	25.4 (5.0–68.8)	90.9 (88.9–92.6)	87.2 (75.7–94.6)	
	IC	9/13 (69.2), (38.6–90.9)	50/51 (98.0), (89.6–100)	79.7 (35.3–96.6)	96.6 (92.7–98.5)	95.2 (86.7–99.0)	
Pt/Et inhA	CS	4/11 (36.4), (10.9–69.2)	9/10 (90.0), (55.5–99.8)	28.8 (5.1–75.2)	92.7 (88.6–95.4)	84.6 (62.4–96.5)	
	IC	11/16 (68.8), (41.3–89.0)	10/12 (83.3), (51.6–97.9)	31.4 (11.0–62.9)	96.0 (91.8–98.1)	81.9 (62.8–93.8)	
Lfx <i>gyrA</i>	CS	2/23 (8.7), (1.1–28.0)	40/43 (93.0), (80.9–98.5)	12.2 (2.4–43.5)	90.2 (88.8–91.4)	84.6 (73.6–92.3)	
	IC	14/24 (58.3), (36.6–77.9)	38/41 (92.7), (80.1–98.5)	47.0 (22.9–72.1)	95.2 (84.9–99.3)	89.3 (79.1–95.6)	
Mfx _{1.0} b <i>gyrA</i>	CS	1/17 (5.9), (0.2–28.7)	45/49 (91.8), (80.4–97.7)	7.4 (0-60.3)	89.8 (79.3–96.1)	83.2 (72.0–91.3)	
	IC	10/24 (41.7), (22.1–63.4)	33/40 (82.5), (67.2–92.7)	20.9 (5.4–47.1)	92.7 (81.2–98.3)	78.4 (66.4–87.7)	
Mfx _{0.25} gyrA	CS	2/16 (12.5), (1.6–38.4)	15/16 (93.8), (69.8–99.8)	18.2 (0.03-83.2)	90.6 (73.9–98.2)	85.6 (68.7–95.5)	
	IC	5/7 (71.4), (29.0–96.3)	5/6 (83.3), (35.9–99.6)	32.3 (4.0–77.0)	96.3 (54.1–100)	82.1 (51.8–97.2)	

CIs for sensitivity, specificity, and accuracy are 'exact' Clopper-Pearson CIs.

Positive Predictive Value (PPV), Negative Predictive Value (NPV) and Accuracy are dependent on disease prevalence.

^bCC value of 0.25 mg/L and clinical breakpoint of 1.0 mg/L was used for MGIT; Mfx CC of 1 mg/L for LJ was set.

CC, critical concentration; CI, confidence interval; Cm, capreomycin; CS, clinical specimen (mostly sputum and one sample of pleural fluid); E, ethambutol; H, isoniazid; IC, culture isolate; Km, kanamycin; Lfx, levofloxacin; LJ, Löwenstein–Jensen solid egg-based medium; LPA, line probe assay; Mfx, moxifloxacin; Mfx_{0.25}, low-level moxifloxacin resistance with critical concentration value of 0.25 mg/L; Mfx_{1.0}, high-level moxifloxacin resistance with clinical breakpoint of 1.0 mg/L; MGIT, Mycobacteria Growth Indicator Tube; NPV, negative predictive value; PPV, positive predictive value; Pt/Et, prothionamide/ethionamide; R, rifampicin; S, streptomycin; TB, tuberculosis.





Error bars represent 95% confidence intervals.

Cm, capreomycin; E, ethambutol; Et/Pt, ethionamide/prothionamide; H, isoniazid; Km, kanamycin; Lfx, levofloxacin; Mfx_{0.25}, low-level moxifloxacin resistance with critical concentration value of 0.25 mg/L; Mfx_{1.0}, high-level moxifloxacin resistance with clinical breakpoint of 1.0 mg/L; R, rifampicin; S, streptomycin.

LPA testing was performed at the Genetic Research Laboratory (with 7 clinical specimens and 6 culture isolates available for the LPA testing). We found that LPA could correctly exclude the six RR-TB patients transferred to long-term treatment from a shorter regimen, while the remaining single patient with RR-TB was correctly selected for shorter treatment (no further resistance detected by LPA). In five culture isolates and one sputum sample from those excluded patients, at least a mutation in the katG gene was identified, indicating that a high dose of H is likely to be ineffective. Regarding results in clinical samples, four out of seven (57.1%) LPA results were consistent with pDST results. In contrast, two samples had indeterminate results (missing the chance to detect resistance to H). WT bands were detected in regions corresponding to the resistance to H, E, and Pt on pDST in one clinical sample. In three patients who had been excluded from a shorter treatment regimen, one sample showed inferred mutations to H and Pt. In one specimen showed inferred resistance to H, and embB M306V mutations were detected, and one sample showed katG, embB M306V mutations, and inferred resistance to FQ. Finally, the patient with RR-TB who continued the shorter treatment had WT results (in the regions corresponding to other drugs) detected by LPA and no additional resistance detected in pDST.

We have also calculated TAT for first- and second-line pDST in days from sputum collection to reporting of results. All LPA results were available during the second part of the next working day after sample shipment to the laboratory (max TAT was 2 days). In our cohort of patients, the TAT of pDST for H was 30.5 (IOR 36) days, for R - 30.5 (IQR 37) days, for E - 30.0 (IQR 36) days, for S - 30.0 (IQR 30) days, for Km and Cm - 38.5 (IQR 48) days, for Lfx - 37.5 (IQR 48) days, for Mfx - 38.0 (IQR 48) days, and for Pt/Et - 71.0 (IQR 49) days. Implementation of LPA has shown a 15-fold (30/2) to 36-fold (71/2) reduction in the time needed for DST. It could also reduce the triage time for suitable ward sharing by infection control requirements and time under exposure to anti-TB drugs with undetected resistance.

Discussion

This study provides evidence of high diagnostic accuracy of LPA (AID Autoimmun Diagnostika GmbH) for detecting resistance to core anti-TB drugs such as H (80.1%, 95% CI 68.5–89.0), R (74.7%, 95% CI 62.4–84.6), and FQ (e.g. Mfx_{0.25} – 85.6%, 95% CI 68.7–95.5) in clinical samples. We found that initial LPA testing in sputum specimens could correctly guide clinicians in choosing between longer and shorter TB treatment

Table 2. Comparison of LPA tests for detection of resistance to anti-TB drugs in clinical samples and culture isolates, with phenotypic drug susceptibility testing as reference standard.

Anti-TB	Sensitivity (%), (95% Cl)		z Test,	Specificity (%), (95% CI)	z test, two-	
arug	Clinical samples	Culture isolates	p value	Clinical samples	Culture isolates	tailed <i>p</i> value
Н	23/30 (76.7), (57.7–90.1)	30/30 (100), (88.4–100)	0.005	3/4 (75.0), (19.4–99.4)	4/4 (100), (39.8–100)	0.285
R	16/29 (55.2), (35.7–73.6)	24/29 (82.8), (64.2–94.2)	0.023	5/5 (100), (47.8–100)	5/5 (100), (47.8–100)	1
E	6/19 (31.6), (12.6–56.6)	8/19 (42.1), (20.3–66.5)	0.503	11/12 (91.7), (61.5–99.8)	11/12 (91.7), (61.5–99.8)	1
S	2/3 (66.7), (9.4–99.2)	3/3 (100), (29.2–100)	0.276	3/4 (75.0), (19.4–99.4)	4/4 (100), (39.8–100)	0.285
Km	2/15 (13.3), (1.7–40.5)	5/15 (33.3), (11.8–61.6)	0.194	18/19 (94.7), (74.0–99.9)	18/19 (94.7), (74.0-99.9)	1
Cm	2/10 (20.0), (2.5–55.6)	5/10 (50.0), (18.7–81.3)	0.159	22/23 (95.7), (78.1–99.9)	22/23 (95.7), (78.1–99.9)	1
Pt/Et	4/9 (44.4), (13.7–78.8)	7/9 (77.8), (40.0–97.2)	0.147	2/2 (100), (15.8–100)	2/2 (100), (15.8–100)	1
Lfx	1/12 (8.3), (0.2–38.5)	6/12 (50.0), (21.1–78.9)	0.024	15/16 (93.8), (69.8–99.8)	14/16 (87.5), (61.7–98.5)	0.542
Mfx _{1.0}	1/11 (9.1), (0.2–41.3)	5/11 (45.5), (16.8–76.6)	0.055	16/17 (94.1), (71.3–99.9)	15/17 (88.2), (63.6–98.5)	0.542
Mfx _{0.25}	1/6 (16.7), (0.4–64.1)	4/6 (66.7), (22.3–95.7)	0.078	4/4 (100), (39.8–100)	3/4 (75.0), (19.4–99.4)	0.285

We used *z* test to compare two proportions. Comparative performance of LPA was analyzed between 2 smear microscopy negative (SM–) and 32 smear-positive (SM+) clinical samples *versus* 34 culture isolates in the same patients for H, R, Km; 11 SM+ clinical samples *versus* 11 culture isolates for Pt/Et; 2 smear microscopy negative (SM–) and 31 smear-positive (SM+) clinical samples *versus* 33 culture isolates for Cm; 2 smear microscopy negative (SM–) and 29 smear-positive (SM+) clinical samples *versus* 31 culture isolates for E; 1 SM– and 6 SM+ clinical samples *versus* 7 culture isolates for S, 1 SM– and 27 SM+ clinical samples *versus* 28 culture isolates for Lfx and Mfx₁₀; 1 SM– and 9 SM+ clinical samples *versus* 10 culture isolates for Mfx_{0.25}.

Cl, confidence interval; Cm, capreomycin; E, ethambutol; H, isoniazid; Km, kanamycin; Lfx, levofloxacin; LPA, line probe assay; Mfx_{0.25}, low-level moxifloxacin resistance with critical concentration value of 0.25 mg/L; Mfx_{1.0}, high-level moxifloxacin resistance with clinical breakpoint of 1.0 mg/L; Pt/Et, prothionamide/ethionamide; R, rifampicin; S, streptomycin; TB, tuberculosis.

regimens in 57.1% of RR-TB cases. However, the WHO recommends minimal requirements to target product profiles for the next generation of the gDST at peripheral centers, such as aggregated sensitivity of at least 88% for all clinically relevant mutations and specificity $\geq 98\%$,²⁶ which could only be obtained for culture isolates tested in the diagnostic module that identifies resistance to R, H, and Pt/Et. A within-study comparison of LPA confirmed (two-tailed z test) that using culture isolates was associated with higher resistance detection accuracy than sputum samples (except for FQ). Furthermore, a combination of rpoB, katG, and inhA mutations increased the probability of FQ resistance by 3.8 times. It can be considered an 'off-label' alternative for the treatment decision-making process in resource-limited settings (e.g. irregular supply of reagents/consumables for molecular or culture-based DST for FQ) or when gDST to FQ has suboptimal sensitivity.

Initial exclusion of FQ resistance is a decision-making step for the recently recommended shorter regimen for drug-susceptible TB (2months isoniazid (H)-rifapentine (P)-pyrazinamide (Z)-moxifloxacin

(Mfx) followed by 2 months HP-Mfx)²⁷ and drugresistant TB schemes, including Hr-TB with 6 months RZE-Lfx and RR/MDR-TB with 6-month BPaLM/BPaL regimens.28 However, the low sensitivity of this assay for identifying FQ resistance could be associated either with using only the gyrA locus as a quinolone-resistance determining region or with a prevalence of unknown mutations in dominant M. tuberculosis strain types circulating among our cohort of patients (atypical mutations outside of the interrogated regions of the test). Moreover, systematic errors of LPA are possible due to synonymous and non-synonymous mutations in the gyrA region of some M. tuberculosis isolates, preventing the binding of WT and mutant probes.²⁹ In another commercial LPA evaluation study, 63% of sputum sample results were classified as resistant based only on the lack of hybridization with the WT probe associated with a double gyrA mutation, 80Ala and 90Gly, detected by gyrA sequencing.30

Although culture-based DST methods remain the gold standard for identifying drug resistance, these tests require a consolidated approach to select optimal CC, allowing minimization of false-susceptible and false-resistant results with low-resistant strains containing a gyrA mutation.³¹ Taking into account the diagnostic importance of pDST on solid media in resource-limited settings, we have also included an interim CC for Mfx of 1 mg/L on Löwenstein-Jensen solid eggbased medium (LJ) in our analysis. However, it could lead to misclassification of gyrA mutants of M. tuberculosis as a susceptible strain at this concentration due to disparities in the validation of this CC with other commercial LPAs.³² Predominantly M/XDR M. tuberculosis Beijing clades circulating in our region could contribute to individually acquired mutations associated with additional drug resistance.³³ After that, a better understanding of the genetic diversities of M. tuberculosis strains can help to adapt diagnostic algorithms according to each country's actual needs.

To avoid pseudo-outbreaks or minimize underestimation of the risk of resistance, TB clinical management teams should follow evidence-based practices in the interpretation of gDST results. Conceptually, the WT strain is a drug-sensitive variant, but absence of the WT probe is not equal to the detection of the mutation. However, in an earlier expert report³⁴ on this issue, the sensitivity and specificity calculation methodology could have been better described, or a better approach used when an absent WT probe has been considered documented resistance, which is discordant with current recommendations.²² LPA-based detection of resistance to anti-TB drugs is focused on the hotspot regions due to limited space on the strip. However, the results of such testing give a wider pattern for interpretation: mutation detected (MUT probe band developed conferring resistance), heteroresistance (WT and MUT band coexisting at the same loci), all WT probes are developed without mutation band (resistance not detected) and no WT band developed (resistance inferred). Nonetheless, discrepancies between LPA and pDST are better explained through clinical implications or additionally needed diagnostic actions instead of considering the data as inconsistent results.¹⁰ We represented the rate of uninterpretable LPA results as a proportion of all tests performed by each module irrespectively of pDST result availability because, for some drugs such as S, Pt/Et, and FO, we have performed many more LPA tests than there are available results of pDST for **Table 3.** Mutations detected by line probe assay in clinical samples and culture isolates from patients with GeneXpert positive for *Mycobacterium tuberculosis*, but negative for rifampicin detection.

Mutations detected	Ν
Clinical samples, 6 out of 17 (35.5%)	
katG S315T, rpsL K43R, rpoB S531L, rpoB S531W	1
<i>rpoB</i> S531T	1
rrs a1401g, rrs c513t, rpsL K88R	1
katG S315T, inhA, rpoB S531L, S531W	1
katG S315T, rpsL K43R	1
rpoB S531L	1
Culture isolates, 14 out of 35 (40%)	
katG S315T, rpsL K43R	3
katG S315T	1
rpsL K43R	1
katG S315T, rpoB S531L, embB M306V	1
katG S315T, inhA, rpoB S531L, rrs a1401g, rpsL K43R, gyr S91P, embB M306V	A 1
rpsL K88R	1
katG S315T, inhA, rpoB s531L, gyrA S91P, embB M306V	1
katG S315T, rpsL K88R	1
katG S315T, rpoB S531L, rpoB S531W	1
<i>katG</i> S315T, <i>rpsL</i> K88R, <i>rpsL</i> K43R, <i>rrs</i> a514c	1
katG 315 S315T embB M306V, rpoB S531L, rpoB S531W	1
inhA	1

comparison (e.g. patients with susceptible TB had results of LPA but rarely pDST was done for second-line drugs). The rate of inconclusive results was significantly higher while analyzing clinical samples (p < 0.05 for all modules). To explain conceptual differences in determining resistance between phenotypical and molecular-based DST, we described a detailed distribution of the results (including indeterminate) for both methods in Tables 4 and 5.

Our findings show that in over 35% of clinical samples from patients without the *rpoB* mutation detected by Xpert, LPA identified mutations in

THERAPEUTIC ADVANCES in *Respiratory Disease*

Drug	Specimen type	Resistance det	ected, n [%]	Resistance not det	ected, n [%]	Not evaluated, n {%}	Indeterminate, n [%]
		MUT probe band developed	Heteroresistance (WT and MUT band coexisting at the same gene target)	Resistance inferred (no WT band developed)	All WT probes are developed, and no mutation band	No pDST available	Proportion to all tests performed by each module irrespectively to pDST result availability
т	Clinical sample	41 (62.1)	3 (4.5)	13 [19.7]	9 (13.6)	I	16/82 (19.5) Clinical samples
	Culture isolate	69 (80.2)	2 (2.3)	1 [1.2]	14 [16.3]	I	
Ъ	Clinical sample	1 (1.5)	35 (53.0)	19 [28.8]	11 [16.7]	I	
	Culture isolate	4 (4.7)	53 (61.6)	9 [10.5]	20 (23.3)	I	
Pt/Et	Clinical sample	4 [19.0]	1 (4.8)	11 [52.4]	5 (23.8)	45	
	Culture isolate	13 (46.4)	1	3 [10.7]	12 (42.9)	58	3/89 [3.4] Culture isolates $p = 0.0008$
ш	Clinical sample	8 [11.4]	5 (7.1)	21 (30.0)	36 [51.4]	6	8/84 (10.3) Clinical samples
	Culture isolate	25 (30.9)	1 (1.2)	I	54 (66.7)	I	
Lfx	Clinical sample	5 [7.6]	I	60 [90.9]	1 [1.5]	10	
	Culture isolate	17 [26.2]	I	27 (41.5)	21 (32.3)	15	
Mfx _{0.25}	Clinical sample	3 [9.4]	I	29 (90.6)	I	50	
	Culture isolate	6 [46.2]	ı	5 (38.5)	2 [15.4]	67	
Mfx _{1.0}	Clinical sample	5 [7.6]	ı	60 (90.9)	1 (1.5)	10	p=0.01278
	Culture isolate	17 [26.6]	I	20 (31.3)	27 (42.2)	16	
S	Clinical sample	I	7 (50.0)	2 [14.3]	5 (35.7)	58	12/84 (14.3) Clinical samples
	Culture isolate	5 [18.5]	10 (37.0)	I	12 (44.4)	55	
Кm	Clinical sample	1 [1.7]	3 (5.0)	16 [26.7]	40 (66.7)	12	
	Culture isolate	7 [10.6]	3 (4.5)	2 (3.0)	54 [81.8]	16	
Cm	Clinical sample	1 [1.8]	3 (5.3)	15 [26.3]	38 (66.7)	15	
	Culture isolate	6 [9.4]	4 (6.3)	2 (3.1)	52 (81.3)	18	4/86 (4.7) Culture isolates $p=0.03236$
Cm, capre with clinice	omycin; E, ethambutol; H, isonia: al breakpoint of 1.0 mg/L; MUT, n	zid; Km, kanamycin; nutation probe; pDS	Lfx, levofloxacin; Mfx _{0.25} , low-level T, phenotypic drug susceptibility te	. moxifloxacin resistanc est; Pt/Et, prothionamic	e with critical concentra le/ethionamide; R, rifam	tion value of 0.25 mg picin; S, streptomyci	/L; Mfx _{1.0} , high-level moxifloxacin resistance n; WT, wild-type probe.

Volume 18

Anti-TB drugs	Resistant, <i>n</i> (%)		Susceptible, n (%)		Indeterminate	e,ª n (%)
	MGIT	LJ	MGIT	LJ	MGIT	LJ
Н	72 (60.0)	21 (17.5)	15 (12.5)	9 (7.5)	2 (1.7)	1 (0.8)
R	73 (57.9)	18 (14.3)	20 (15.9)	12 (9.5)	2 (1.6)	1 (0.8)
E	46 (41.1)	13 (11.6)	38 (33.9)	12 (10.7)	2 (1.8)	1 (0.9)
S	9 (23.7)	9 (23.7)	14 (36.8)	3 (7.9)	2 (5.3)	1 (2.6)
Km	30 (30.3)	9 (9.1)	40 (40.4)	17 (17.2)	2 (2.0)	1 (1.0)
Cm	15 (16.0)	4 (4.3)	53 (56.4)	19 (20.2)	2 (2.1)	1 (1.1)
Lfx	30 (30.0)	8 (8.0)	42 (42.0)	17 (17.0)	2 (2.0)	1 (1.0)
Mfx _{1.0}	18 (18.2)	8 (8.1)	53 (53.5)	17 (17.2)	2 (2.0)	1 (1.0)
Mfx _{0.25}	17 (45.9)	-	18 (48.6)	-	2 (5.4)	-
Pt/Et	12 (27.9)	7 (16.3)	8 (18.6)	13 (30.2)	2 (4.7)	1 (2.3)

Table 5. Phenotypic drug susceptibility test results for each anti-TB drug evaluated in the study.

^aContamination and uninterpretable results.

Cm, capreomycin; E, ethambutol; H, isoniazid; Km, kanamycin; Lfx, levofloxacin; LJ, Löwenstein–Jensen solid eggbased medium; Mfx_{0.25}, low-level moxifloxacin resistance with critical concentration value of 0.25 mg/L; Mfx_{1.0}, highlevel moxifloxacin resistance with clinical breakpoint of 1.0 mg/L; MGIT, Mycobacteria Growth Indicator Tube; Pt/Et, prothionamide/ethionamide; R, rifampicin; S, streptomycin; TB, tuberculosis.

other regions that could be associated with resistance to anti-TB drugs. In areas with a high prevalence of drug resistance and limited capacity for single-occupancy patient-care rooms, like Ukraine, it is essential to avoid the prescription of standardized treatment regimens for RR-TB patients based on XpertMTB/RIF testing only, as it may lead to nosocomial transmission of mutant *M. tuberculosis* strains. On the other hand, the current approach, where only patients with Xpert-detected *rpoB* mutations remain eligible for second-line pDST testing, could contribute to an underestimation of resistance due to selection bias.

Reprioritization of resources due to COVID-19 (including exploitation of the Xpert platform to perform molecular detection of SARS-CoV-2) throughout Ukraine might have worsened the drug-resistant TB delay during the pandemic's peak. In a recent cross-sectional multicenter study, Penn-Nicholson *et al.*³⁵ found that detection of H, FQ, Et, and SLIs resistance by the Xpert MTB/XDR assay and by LPA had equivalent accuracy. Black *et al.*³⁶ in a retrospective laboratory-based study, concluded that discordance between XpertMTB/RIF and other commercial LPA (MTBDRplus) results could be associated with false detection of RIF-resistance by Xpert (22%), heteroresistance of samples (16%) and some errors in either transcription (7%) or interpretation of LPA results (7%). In our study, lower sensitivity of LPA on clinical specimens to detect R resistance (64.7 versus 80.4) compared to H could be explained by a relatively more comprehensive range of genetic loci associated with resistance to H than to R.37 However, in our cohort of patients, using XpertMTB/RIF instead of pDST as a reference standard test, detection of RR-TB by LPA was associated with even lower sensitivity 61.1 (95% CI 46.9-74.1) versus 64.7 (95% CI 50.1–77.6), p = 0.704 and specificity 75.0 (95%) CI 42.8–94.5) versus 80.0 (51.9–95.7), p=0.757, respectively. It could also be influenced by the lower limit of detection for XpertMTB/ RIF (112.6 CFU/mL)¹¹ compared with LPA (10,000 CFU/mL).³⁸ On the other hand, low TB bacillary load in clinical samples could increase the probability of false R resistance detection by Xpert assay, while CC used in pDST can misclassify strains with specific R resistance-conferring mutations located at codon 526, 511Pro, 516Tyr, 533Pro, and 572Phe of the rpoB gene.³⁹

About half of the samples tested for R and S have heteroresistance in our cohort. There are three ways heteroresistance can be explained; first, it is seen in coexisting susceptible and resistant bacterial cells due to naturally developing spontaneous mutations in the initial population. The second situation is when a patient is infected/re-infected with two different M. tuberculosis strains or acquires resistance during exposure to ineffective treatment. Finally, it might also be caused by different limits of detection for WT and MUT probe reactions for R and S compared with the rest of probes.⁴⁰ As a result of the investigation of over 31,000 MTBC genomes, Vargas Ir et al.41 demonstrated that the presence of loss-of-function mutations in the coding region of eis can lead to systematic diagnostic errors of many molecular assays to detect resistance to SLIs.

A low proportion of patients (5%) in the RR-TB cohort who completed the shorter regimen with an injectable agent (known as the 'Bangladesh' regimen) was consistent with estimations of other studies.42,43 Our findings could apply to a new 9-month entirely oral regimen containing Bdg (instead of an injectable agent) (4-6 Bdq-Mfx/ Lfx-Pt/Et-Cfz-Z-H_{hd}-E/5Mfx-Cfz-Z-E) for RR/ MDR-TB. However, in our study, adding a single new component (Bdg) is unlikely to be sufficient to cover a high rate of resistance (63%) and intolerance (14%) to other documented components of this regimen. Moreover, using Bdq in these circumstances may create an additional risk of acquiring resistance to this important drug. Considering the abovementioned precautions, we suggest replacing Pt/Et with 2 months of Lzd (600 mg daily) whenever possible during the initial phase of the short regimen for an RR/ MDR-TB cohort of patients in Ukraine. High dose H (10-15 mg/kg per day in adults) is likely effective if susceptibility is confirmed or only specific inhA promoter mutations are detected (without katG mutations).44 However, this approach neglects the prediction of resistance to Pt/Et, making prescribing high dose H controversial. Due to high probability of preXDR-TB in patients with documented combination of katG, rpoB, and inhA mutations, 6-month BPaL regimen (without Mfx) could be more suitable in such population living in our geographic area.

Our study has some limitations. We did not have the possibility to analyze the discordance

between genotypic and pDST by whole genome sequencing of the M. tuberculosis isolates. TAT for pDST results depends on the time to culture positivity, and it is significantly influenced by media (MGIT versus LJ) and bacterial load in sputum specimens, which should be considered for the pragmatic evaluation of LPA benefits. Critical concentrations of some anti-TB drugs have not been updated for many years, affecting the design and interpretation of molecular assays. LPA is a moderate-complexity molecular DST associated with some technical issues, such as open plate format with possible cross-contamination and compromised interpretation if either all bands or none are developed.²² The high proportion 91/126 (72.2%) of RR/MDR-TB patients in the study cohort corresponded to national hospitalization practices with more prevalent initial ambulatory treatment for drugsusceptible TB but, on the other hand, it could potentially impact the accuracy estimates for individual drug resistance detection by LPA. To minimize this effect while using diagnostic test evaluation calculator, we set up the prevalence rate of resistance to first-line medicines (H, R, and E) at 35% and the rate of resistance to second-line drugs (S, Km, Cm, Lfx, Mfx, Pt/Et) at 10%, representing national surveillance data.²⁵ The association between katG, rpoB and inhA mutations and FQ resistance may be setting-specific and therefore the findings should not be extrapolated beyond studied populations without local molecular epidemiology data analysis.

However, our study has several important strengths. As far as we know, our findings were based on the first direct within-study comparison of LPA results between clinical samples and culture isolates from the same cohort of patients. Another strength of our study is the development of local capacity for the implementation of LPA technology in the Genetic Research Laboratory in Vinnytsia, obtaining results in only 1–2 working days, in comparison with the standard procedure of shipping the samples to the reference laboratory located in Kyiv (at 300 km). Another advantage of our research implementation was enrolling patients into the study regardless of smear microscopy status, where one in five had SM-negative results. Herein, the inconclusive rate of the investigated LPA among SM-negative clinical samples varied from 20.0% (3/15) for FQ-E to 43.8%

(7/16) for the MDR-TB module in comparison with over 60% results that could not be interpreted in another commercially available assay.⁴⁵

Conclusions

In summary, accuracy of LPA varied from 71.4% (S) to 87.2% (Cm) in clinical specimens and from 78.4% (Mfx_{1.0}) to 95.2% (H, Cm) in the corresponding culture isolates depending on the anti-TB drugs studied and the type of sample used. Implementing LPA in high M/XDR-TB burden settings can facilitate clinicians providing an optimal distribution of available wards between patients with different profiles of resistance and quick antimicrobial stewardship in the clinical management of TB.

Declarations

Ethics approval and consent to participate

The Bioethics Committee approved the study at National Pirogov Memorial Medical University, Vinnytsia, Ukraine, protocol №5 from 3 July 2018. The study was performed according to the Guidelines for Good Clinical Practice with proper certification of investigators. We did not use any personal identification information for analysis. Each participant gave written informed consent before enrolment in the study. If a participant wanted to withdraw from a study either verbally or in writing, the patient's information was eliminated from the dataset.

Consent for publication

Not applicable.

Author contributions

Andrii Dudnyk: Conceptualization; Data curation; Formal analysis; Investigation; Methodology; Project administration; Resources; Software; Visualization; Writing – original draft, Writing – review & editing.

Matthias Hempel: Investigation; Validation; Writing – review & editing.

Oksana Lytvyniuk: Formal analysis; Investigation; Writing – review & editing.

Halyna Liudkevych: Formal analysis; Investigation; Project administration; Writing – review & editing. **Volodymyr Matsera:** Data curation; Formal analysis; Project administration; Validation; Writing – review & editing.

Tetiana Nikitchenko: Data curation; Formal analysis; Validation; Writing – review & editing.

Svitlana Blyzniuk: Data curation; Formal analysis; Project administration; Validation; Writing – review & editing.

Barbara Molina-Moya: Data curation; Formal analysis; Project administration; Writing – review & editing.

Rosemarie Preyer: Investigation; Resources; Writing – review & editing.

José Domínguez: Conceptualization; Formal analysis; Funding acquisition; Methodology; Project administration; Supervision; Writing – review & editing.

Acknowledgements

We thank the patients included in the study for their cooperation and consent to participate. We also thank microbiology laboratory personnel at the Regional Phthisiopulmonology Center and the Genetic Research Laboratory technician for their contributions to this study.

Funding

The authors disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: This research was supported by (i) a grant from the Instituto de Salud Carlos III (PI19/01408, and PI22/01834), integrated into the Plan Nacional I+D+I and cofunded by the ISCIII Subdirección General de Evaluación and the Fondo Europeo de Desarrollo Regional (FEDER) and CERCA Programme/ Generalitat de Catalunya; (ii) the European Union's Horizon 2020 Research and Innovation Programme under the Marie Skłodowska-Curie grant agreement No. 823854 (INNOVA4TB); (iii) from COST (European Cooperation in Science and Technology) action ADVANCE-TB (CA21164); and (iv) this project has received funding through the MSCA4Ukraine project ID:1233336, which is funded by the European Union.

Competing interests

MH and RP report a relationship with AID/ GenID GmbH, including employment. JD has a patent on nontuberculous mycobacteria immunology licensed to AID/GenID GmbH. The other authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Availability of data and materials

The datasets generated and analyzed during the current study are not publicly available due to restrictions related to the informed consent of the participants and data sharing agreement. Still, they are available from the corresponding author on reasonable request.

ORCID iD

Andrii Dudnyk ២ https://orcid.org/0000-0002-8780-9480

Supplemental material

Supplemental material for this article is available online.

References

- World Health Organization. Global tuberculosis report 2022, World Health Organization, Geneva, Licence: CC BY-NC-SA 3.0 IGO, https://www. who.int/publications/i/item/9789240061729 (2022, accessed 16 November 2023).
- World Health Organization. Catalogue of mutations in *Mycobacterium tuberculosis* complex and their association with drug resistance, WHO | World Health Organization, https://www.who. int/publications/i/item/9789240028173 (2021, accessed 17 October 2023).
- World Health Organization. Rapid communication: key changes to the treatment of drug-resistant tuberculosis, https://www.who. int/publications/i/item/WHO-UCN-TB-2022-2 (2022, accessed 16 September 2023).
- 4. UNHCR Operational Data Portal (ODP). Ukraine situation Flash update #36, UNHCR Regional Bureau for Europe, https://data.unhcr. org/en/documents/details/97261 (2022, accessed 1 September 2023).
- ReliefWeb. Emergency appeal revised operational strategy. Europe | Ukraine and impacted countries crisis, Regional Office for Europe, Head of Disaster, Climate and Crisis Prevention Response and Recovery, https://reliefweb.int/ report/ukraine/europe-ukraine-and-impactedcountries-crisis-emergency-appeal-revised-

operational-strategy (2023, accessed 19 October 2023).

- World Health Organization. Meeting report of the who expert consultation on the definition of extensively drug-resistant tuberculosis.
 WHO, https://www.who.int/publications/i/item/ meeting-report-of-the-who-expert-consultationon-the-definition-of-extensively-drug-resistanttuberculosis (2021, accessed 14 July 2023).
- Cegielski JP, Chan P-C, Lan Z, et al. Aminoglycosides and capreomycin in the treatment of multidrug-resistant tuberculosis: individual patient data meta-analysis of 12 030 patients from 25 countries, 2009–2016. Clin Infect Dis 2021; 73: e3929–e3936.
- European Centre for Disease Prevention and Control, WHO Regional Office for Europe. Tuberculosis surveillance and monitoring in Europe 2023 – 2021 data, European Centre for Disease Prevention and Control and Copenhagen: WHO Regional Office for Europe, Stockholm, https://www.ecdc.europa. eu/sites/default/files/documents/tuberculosissurveillance-monitoring-2023.pdf (2023, accessed 27 February 2024).
- Public Health Center. Analytical surveillance data for Tuberculosis in Ukraine, 2022, Public Health Center, Kyiv, https://www.phc.org.ua/ sites/default/files/users/user90/TB_surveillance_ statistical-information_2022_dovidnyk.pdf (2023, accessed 28 February 2024).
- Domínguez J, Boeree MJ, Cambau E, et al. Clinical implications of molecular drug resistance testing for *Mycobacterium tuberculosis*: a 2023 TBnet/RESIST-TB consensus statement. *Lancet Infect Dis* 2023; 23: e122–e137.
- World Health Organization. WHO consolidated guidelines on tuberculosis: module 3: diagnosis: rapid diagnostics for tuberculosis detection 2021 update, WHO, https://www.who.int/ publications/i/item/9789240029415 (2021, accessed 12 September 2023).
- 12. Molina-Moya B, Lacoma A, Prat C, *et al.* AID TB resistance line probe assay for rapid detection of resistant *Mycobacterium tuberculosis* in clinical samples. *J Infect* 2015; 70: 400–408.
- Wagner K, Preiswerk B, Keller PM, et al. Defining threshold values in microscopic examination and qPCR for optimal performance of line probe assays for resistance detection in Mycobacterium tuberculosis complex. Diagn Microbiol Infect Dis 2019; 95: 159–161.
- 14. Mahomed S, Mlisana K, Cele L, *et al.* Discordant line probe genotypic testing vs culture-based drug

susceptibility phenotypic testing in TB endemic KwaZulu-Natal: impact on bedside clinical decision making. J Clin Tuberc Other Mycobact Dis 2020; 20: 100176.

- 15. Sinha P, Srivastava GN, Tripathi R, et al. Detection of mutations in the rpoB gene of rifampicin-resistant *Mycobacterium tuberculosis* strains inhibiting wild type probe hybridization in the MTBDR plus assay by DNA sequencing directly from clinical specimens. *BMC Microbiol* 2020; 20: 284.
- World Health Organization. WHO treatment guidelines for drug resistant tuberculosis 2016, update, October 2016 revision, https://www.who. int/publications/i/item/9789241549639 (2016, accessed 10 July 2023).
- 17. World Health Organization. WHO consolidated guidelines on tuberculosis: module 4: treatment drug-resistant tuberculosis treatment, 2022 update, World Health Organization, Geneva, Licence: CC BY-NC-SA 3.0 IGO, https://www.who.int/publications/i/item/9789240063129 (2022, accessed 14 September 2023).
- World Health Organization. Rapid communication: key changes to treatment of multidrug- and rifampicin-resistant tuberculosis (MDR/RR-TB). WHO/CDS/TB/2018.18, World Health Organization, https://apps.who.int/iris/ handle/10665/275383 (2018, accessed 10 July 2023).
- South African National Department of Health. Interim clinical guidance for the implementation of injectable-free regimens for rifampicin-resistant tuberculosis in adults, adolescents and children, NDoH, Pretoria, South Africa, https://www. tbonline.info/media/uploads/documents/dr_tb_ clinical_guidelines_for_rsa_september_2018.pdf (2018, accessed 15 July 2023).
- Cohen JF, Korevaar DA, Altman DG, et al. STARD 2015 guidelines for reporting diagnostic accuracy studies: explanation and elaboration. BMJ Open 2016; 6: e012799.
- World Health Organization. Technical manual for drug susceptibility testing of medicines used in the treatment of tuberculosis, World Health Organization, Geneva, Licence: CC BY-NC-SA 3.0 IGO, https://apps.who.int/iris/ handle/10665/275469 (2018, accessed 20 July 2023).
- 22. World Health Organization. Line probe assays for detection of drug-resistant tuberculosis: interpretation and reporting manual for laboratory staff and clinicians, WHO, https://

www.who.int/publications/i/item/9789240046665 (2022, accessed 14 July 2023).

- Schoonjans F. MedCalc's diagnostic test evaluation calculator. MedCalc, https://www. medcalc.org/calc/diagnostic_test.php (accessed 30 January 2024).
- 24. Social Science Statistics. Z score calculator for 2 population proportions, https://www. socscistatistics.com/tests/ztest/default2.aspx (accessed 30 January 2024).
- 25. Public Health Center. TB surveillance statistical information 2019, Ministry of Health of Ukraine, https://www.phc.org.ua/sites/default/files/users/user90/TB_surveillance_statistical-information_2019_dovidnyk.pdf (2022, accessed 17 July 2023).
- World Health Organization. Target product profile for next-generation drug-susceptibility testing at peripheral centres, World Health Organization, Geneva, Licence: CC BY-NC-SA 3.0 IGO, https://www.who.int/publications/i/ item/9789240032361 (2021, accessed 16 July 2023).
- World Health Organization. WHO operational handbook on tuberculosis module 4: treatment – drug-susceptible tuberculosis treatment, World Health Organization, Geneva, https://www.who. int/publications/i/item/9789240050761 (2022, accessed 25 September 2023).
- Nyang'wa B-T, Berry C, Kazounis E, et al. A 24-week, all-oral regimen for rifampin-resistant tuberculosis. N Engl J Med 2022; 387: 2331– 2343.
- 29. Ajileye A, Alvarez N, Merker M, et al. Some synonymous and nonsynonymous gyrA mutations in *Mycobacterium tuberculosis* lead to systematic false-positive fluoroquinolone resistance results with the Hain GenoType MTBDRsl assays. *Antimicrob Agents Chemother* 2017; 61: e02169-16.
- Kaswa MK, Aloni M, Nkuku L, *et al.* Pseudooutbreak of pre-extensively drug-resistant (Pre-XDR) tuberculosis in Kinshasa: collateral damage caused by false detection of fluoroquinolone resistance by GenoType MTBDRsl. *J Clin Microbiol* 2014; 52: 2876–2880.
- 31. Antimycobacterial Susceptibility Testing Group. Updating the approaches to define susceptibility and resistance to anti-tuberculosis agents: implications for diagnosis and treatment. *Eur Respir J* 2022; 59: 2200166.
- 32. World Health Organization. Technical report on critical concentrations for drug susceptibility

Volume 18

testing of medicines used in the treatment of drug-resistant tuberculosis, World Health Organization, Geneva (WHO/CDS/TB/2018.5), Licence: CC BY-NC-SA 3.0 IGO, https:// www.who.int/publications/i/item/WHO-CDS-TB-2018.5 (2018, accessed 23 August 2023).

- Merker M, Nikolaevskaya E, Kohl TA, et al. Multidrug- and extensively drug-resistant Mycobacterium tuberculosis Beijing Clades, Ukraine, 2015. Emerg Infect Dis 2020; 26: 481–490.
- 34. Stop TB Partnership. Molecular line probe assays for rapid screening of patients at risk of multidrug resistant tuberculosis, Expert group report, Stop TB Partnership, https://stoptb.org/assets/ documents/about/cb/meetings/15/2.08-11%20 Rolling%20out%20diagnostics%20in%20the%20 field/2.08-11.2%20Line%20Probe%20Assays.pdf (2008, accessed 14 July 2023).
- 35. Penn-Nicholson A, Georghiou SB, Ciobanu N, et al. Detection of isoniazid, fluoroquinolone, ethionamide, amikacin, kanamycin, and capreomycin resistance by the Xpert MTB/XDR assay: a cross-sectional multicentre diagnostic accuracy study. *Lancet Infect Dis* 2022; 22: 242–249.
- Black M, Da Silva P and Scott L. Rifampicinresistant TB: discordance between Xpert[®] MTB/ RIF and MTBDR plus results. *Int J Tuberc Lung Dis* 2021; 25: 832–838.
- Nathavitharana RR, Cudahy PGT, Schumacher SG, et al. Accuracy of line probe assays for the diagnosis of pulmonary and multidrug-resistant tuberculosis: a systematic review and metaanalysis. Eur Respir J 2017; 49: 1601075.
- Roscigno G. Drug resistant TB and new diagnostics for people living with HIV: emerging

results from FIND, IAS Conference; Catalysing HIV/TB research, http://www.stoptb.org/wg/tb_hiv/assets/documents/DRUGRE~1.PDF (2009, accessed 15 July 2023).

- Rafael L-L, Raquel M-S, Rogelio F-A, et al. Discordant results between genotypic and phenotypic assays (Xpert MTB/RIF vs. BACTEC MGIT 960 system) for detection of RIF-resistant Mycobacterium tuberculosis isolates in a high burden region. Infect Genet Evol 2021; 96: 105142.
- 40. Ye M, Yuan W, Molaeipour L, *et al.* Antibiotic heteroresistance in *Mycobacterium tuberculosis* isolates: a systematic review and meta-analysis. *Ann Clin Microbiol Antimicrob* 2021; 20: 73.
- 41. Vargas R Jr, Freschi L, Spitaleri A, et al. Role of epistasis in amikacin, kanamycin, bedaquiline, and clofazimine resistance in *Mycobacterium tuberculosis* complex. *Antimicrob Agents Chemother* 2021; 65: e0116421.
- 42. Lange C, Duarte R, Fréchet-Jachym M, et al. Limited benefit of the new shorter multidrugresistant tuberculosis regimen in Europe. Am J Respir Crit Care Med 2016; 194: 1029–1031.
- du Cros P, Khamraev A, Tigay Z, et al. Outcomes with a shorter multidrug-resistant tuberculosis regimen from Karakalpakstan, Uzbekistan. ERJ Open Res 2021; 7: 00537-2020.
- Dooley KE, Miyahara S, von Groote-Bidlingmaier F, et al. Early bactericidal activity of different isoniazid doses for drug-resistant tuberculosis (INHindsight): a randomized, openlabel clinical trial. Am J Respir Crit Care Med 2020; 201: 1416–1424.
- 45. Lutchminarain K, Kajee A, Gandhi NR, et al. Performance of the GenoType MTBDRsl in a programmatic setting, South Africa. Int J Tuberc Lung Dis 2022; 26: 426–432.

Visit Sage journals online journals.sagepub.com/ home/tar

Sage journals